Author's response to reviews

Title: A compare between myocardial topical negative pressure levels of -25 mmHg and -50 mmHg in a porcine model

Authors:

Sandra Lindstedt (sandra.lindstedt@skane.se)
Per Paulsson (per.paulsson@skane.se)
Arash Mokhtari (arash.mokhtari@kane.se)
Bodil Gesslein (bodil.gesslein@med.lu.se)
Joanna Hlebowicz (joanna.hlebowicz@skane.se)
Malin Malmsjo (malin.malmsjo@med.lu.se)
Richard Ingemansson (richard.ingemansson@skane.se)

Version: 8 Date: 16 June 2008

Author's response to reviews:

The following formatting changes have been done;
The afflictions have been changed as requested
List of abbreviations -> Abbreviations
Authors' contribution -> Authors' contributions

Kind Regards

Sandra Lindstedt